

Clinical Ophthalmology





ISSN: (Print) (Online) Journal homepage: informahealthcare.com/journals/doph20

Clinical utility of cyclosporine (CsA) ophthalmic emulsion 0.05% for symptomatic relief in people with chronic dry eye: a review of the literature

Michelle K Rhee & Francis S Mah

To cite this article: Michelle K Rhee & Francis S Mah (2017) Clinical utility of cyclosporine (CsA) ophthalmic emulsion 0.05% for symptomatic relief in people with chronic dry eye: a review of the literature, Clinical Ophthalmology, , 1157-1166, DOI: 10.2147/OPTH.S113437

To link to this article: https://doi.org/10.2147/OPTH.S113437

| 9 | © 2017 Rhee and Mah. This work is published and licensed by Dove Medical Press Limited |
|----------------|--|
| | Published online: 21 Jun 2017. |
| | Submit your article to this journal 🗹 |
| dil | Article views: 136 |
| Q ^L | View related articles 🗗 |
| CrossMark | View Crossmark data 🗗 |
| 2 | Citing articles: 1 View citing articles 🗹 |



REVIEW

Clinical utility of cyclosporine (CsA) ophthalmic emulsion 0.05% for symptomatic relief in people with chronic dry eye: a review of the literature

Michelle K Rhee¹ Francis S Mah²

¹Department of Ophthalmology, Icahn School of Medicine at Mount Sinai, New York, NY, USA; ²Scripps Clinic Torrey Pines, La Jolla, CA, USA **Abstract:** To review the literature on the efficacy of cyclosporine (CsA) ophthalmic emulsion 0.05% on symptomatic relief in chronic dry eye disease. There is consistent evidence of objective improvements in chronic dry eye disease (Schirmer score, corneal and interpalpebral dye staining, and tear breakup time) with CsA, but variable results with symptomatic improvement, possibly due to patient tolerance of CsA, similar comforting effect with artificial tears and CsA vehicle, and the inherent subjective nature of symptom monitoring and analysis. This review explores the literature on CsA with special attention to symptomatic relief.

Keywords: dry eye, cyclosporine, symptoms

Introduction

In 2003, cyclosporine (CsA) ophthalmic emulsion 0.05% was the first Food and Drug Administration (FDA)-approved prescription medication (Restasis®, Allergan, Irvine, CA, USA) for dry eye disease (DED), as well as the first to modify disease¹ rather than to act as a palliative measure as lubricants do. It achieved the indication for increasing tear production in patients with diminished tear production due to ocular inflammation associated with keratoconjunctivitis sicca.² Given the recent FDA approval of lifitegrast ophthalmic solution 5% (Xiidra®, Shire, Lexington, MA, USA), revisiting the literature on CsA is timely. Although the mechanisms of action are different, both CsA and lifitegrast address the underlying inflammatory process in DED.¹¹³

Epidemiology

DED is a widespread and challenging disorder to manage, largely due to its multifactorial causes, chronic nature, need for patient compliance, and limited diagnostic and treatment options. It is the most common form of chronic ocular surface disease;⁴ 5%–34% of the global population is affected.⁵ Forty-three percent of asymptomatic patients have shown signs of dry eye,⁶ indicating a need for earlier diagnosis and intervention to prevent advanced disease. The signs and symptoms of DED often do not correlate,⁷ and there is no gold standard for its diagnosis. Recently, Vehof et al⁸ performed the largest clinical study of DED to date, which identified predictors of discordance between symptoms and signs of DED. Significant predictors of greater symptoms than signs included the presence of a chronic pain syndrome, atopic disease, depression, and osteoarthritis; predictors of lesser symptoms than signs included increased age, primary Sjögren's disease, and graft-versus-host disease (GVHD).

Correspondence: Michelle K Rhee Icahn School of Medicine at Mount Sinai, I Gustave L. Levy Place Box 1183 New York, NY 10029 USA Tel +1 212 241 0204 Email michelle.rhee@mssm.edu

A subset of patients with incongruous signs and symptoms may have neuropathic ocular pain, the result of somatosensory dysfunction and central sensitization, which consists of pain without ongoing peripheral pathology. 10

DED is a significant public health issue. More than 20 million Americans have DED.¹¹ As the aging population rises, so does DED, with more than 15% of those over 65 suffering from it.¹² The Beaver Dam Eye Study found prevalence rates of 8% in those younger than 60 years of age, 19% in those older than 80 years of age, and higher numbers among women than men.¹³ Patients with severe dry eyehad health-related quality-of-life scores in the range of conditions such as class III/IV angina.¹⁴

Patients with dry eye experience symptoms of ocular discomfort, dryness, and episodic visual disturbances. 15–17 The disease is not only uncomfortable but interferes with the ability to work and carry out daily functions. 14,18,19 Although it is not usually a blinding disease, these patients spend a substantial amount of money per year to ease the distress associated with it. Estimated global sales of artificial tears exceeded US\$540 million annually in 2002. 20 Galor et al²¹ analyzed spending data of patients with dry eye and found that in 2005, the mean prescription medication expenditure per patient per year was \$299 in the USA. According to the US Medical Expenditure Panel Survey, glaucoma patients spent a mean of \$556 per year on glaucoma prescription medications in 2006, 22 and spine patients spent a mean of \$397 per year on prescription medications in 2006.

Inflammation in dry eye

Dry eye is a complex disorder of the tears and ocular surface that leads to symptoms of discomfort, visual disturbance, tear film instability, and potential damage to the ocular surface. Increased osmolarity of the tear film and inflammation of the ocular surface are integral to the disease.¹¹

Inflammation of the ocular surface and lacrimal gland can occur as both the instigator and product of DED. The understanding of inflammation as a key factor in DED supports the use of topical corticosteroids and topical CsA. This was a therapeutic turning point; the underlying mechanism of pathology was targeted rather than mollifying symptoms with the conventional strategy of lubrication.

Management of dry eye

An arsenal of treatment options exists for DED.²⁴ They include the following:

 Lubricants, including artificial tears, gels, ointments, inserts.

- 2. Anti-inflammatory agents, such as topical CsA, corticosteroids, lifitegrast, essential fatty acids, and oral tetracyclines.
- 3. Environmental and behavioral modifications, such as the use of humidifer, purposeful blinking, and computer screen adjustment.
- 4. Cessation of systemic medications linked to DED, such as antihistamines and other anticholinergic agents.
- 5. Others including punctal occlusion, oral secretogogues, pulse corticosteroids, autologous serum, mucolytic therapy, moisture chamber spectacles, management of eyelids, contact lens (CL) therapy, and acupuncture.²⁵

While topical corticosteroids are effective in breaking the cycle of inflammation, their known side effects, such as ocular hypertension, cataract, decreased wound healing, and predisposition to infection limit chronic use. ²⁶ Alternatively, topical CsA has a favorable risk—benefit profile for chronic use. Blood levels of CsA are barely detectable at a maximum level of 0.16 ng/mL, ²⁷ and the most common side effect of CsA is ocular burning. ²⁸ Other side effects of CsA include blurred vision, ocular itching, conjunctival hyperemia, discharge, foreign body sensation, and stinging. ²⁹

CsA is a lipophilic, neutral cyclic undecapeptide consisting of 11 amino acids.³⁰ It acts as an immunosuppressant by inhibiting T-cell-mediated inflammation and cytokines in the conjunctiva,^{1,31} stimulates natural tear production,³² and increases goblet cell density.³³ Topical CsA enhances ocular surface health.

Methods

A PubMed search was performed using the keywords "cyclosporine dry eye," "cyclosporine ophthalmic and dry eye," and "topical cyclosporine and thyroid eye disease." All searches were limited to papers published in or translated into the English language, ranging from 1986 to March 2017. The papers reviewed were limited to human studies that focused on symptomatic relief of chronic dry eye with CsA emulsion 0.05% treatment. There were no restrictions on the study design.

Results

CsA for moderate-to-severe dry eye

When considering objective signs such as Schirmer scores and ocular staining, CsA 0.05% quite reliably outperformed vehicle and artificial tears in several randomized controlled trials.^{27,32,34–38} However, despite many groups^{27,29,32,34,36,39–43} reporting beneficial effects of CsA on symptomatic relief in DED, it is challenging to compare and produce consistent

conclusions given the various study designs. Nevertheless, in general, CsA was found to improve at least one symptom, usually ocular dryness, with variability among studies regarding ocular surface disease index⁴⁴ (OSDI) and particular symptom alleviation (Table 1).

As part of The Cyclosporin A Phase II Study Group, Stevenson et al²⁷ led a randomized, multicenter, double-masked controlled trial of 162 patients with either Sjögren's syndrome (SS) or non-SS dry eye. Hundred twenty-nine patients received CsA (varying concentrations of 0.05%, 0.1%, 0.2%, and 0.4%), and 33 patients received vehicle. CsA 0.05% produced the most consistent improvement in patient symptoms such as ocular dryness and sandy/gritty feeling. However, OSDI scores differed in that at treatment week 12, only the 0.1% and 0.2% groups had significant improvement, and at 4 weeks posttreatment, only the 0.2% group had sustained significant improvement; there was no change or worsening in the vehicle group.

The subsequent Cyclosporin A Phase III Study Group³² performed two identical multicenter, randomized, double-masked, parallel-group clinical trials focusing on the efficacy of CsA 0.05% and 0.1% to vehicle in 671 patients with moderate-to-severe DED with or without SS. Blurred vision was the most prominent subjective change in the CsA 0.05% group compared to vehicle at all follow-up visits, which ended at 6 months. This is consistent with the Phase II Study Group finding that the 0.05% group had the most reliable subjective improvement, although it was in ocular dryness and sandy/gritty feeling. In addition, the 0.05% group required less rescue artificial tear drops. With OSDI, again, there was no significant difference among the groups.

Rao³⁷ performed a single-center, investigator-masked, prospective randomized trial comparing CsA 0.05% (36 patients) to vehicle (Refresh Endura®, Allergan; 22 patients) for 12 months in moderate-to-severe DED. OSDI scores were significantly better in the CsA group at months 8 and 12 when compared to the vehicle group. Likewise, Chen et al³⁴ studied 233 Chinese patients with moderate-to-severe DED in a multicenter, randomized, double-masked 8-week trial comparing CsA 0.05% to vehicle. They found CsA superiority in symptomatic relief, as measured by ocular dryness and foreign body sensation.

In a large patient survey study⁴⁵ of 5,884 patients, recruited by 4,504 ophthalmologists, optometrists, and primary care physicians throughout the USA, CsA 0.05% was found to significantly improve symptom severity by 30% and activity impairment by 31%–36%. The greatest impact on symptom reduction and activity impairment occurred

after 30 days, with continued reduction up to the 60-day follow-up. This survey noted a very rapid response in symptomatic relief; 41% experienced symptomatic relief within 1–3 weeks. This is in contrast to other studies that required 6 months to achieve statistical significance for improvement in symptoms.^{27,32} Although this study had limitations with regard to lack of study protocol, lack of objective clinical testing, and dependence on patient self-reported subjective information, the large volume of patients evaluated in a nonclinical trial setting nonetheless added observational information, corroborating clinical trial results.

An important rationale for determined encouragement of CsA therapy is its ability to reduce disease progression,³⁷ and progressive improvements in corneal staining with treatment maintained over 24 months. 46 Wilson and Perry 47 showed that CsA could prevent progression of DED in some patients. Although it was a small retrospective case series, they presented 8 patients (3.9% of patients with dry eye from one practice, and 1.5% from another practice) with chronic DED who were essentially cured (free of symptoms and signs of disease) for at least 1 year after completion of CsA treatment. Perry et al²⁹ prospectively evaluated 158 patients who were unresponsive to artificial tears, and monitored OSDI for symptomatic improvement. They found that 74.1% of mild DED, 72.4% of moderate DED, and 66.7% of severe DED improved with CsA bid. The new information gleaned from this study was that the biggest improvement in OSDI occurred in those with mild disease, suggesting that early treatment of DED may produce the best results, including decelerating disease progression.³⁷

CsA and topical steroids

Barriers to patient tolerance and acceptance of CsA therapy are ocular burning (10.9%), ocular stinging (3.9%), and conjunctival hyperemia (3.4%).²⁸ These symptoms impact patient tolerance and adherence to medication, leading to dropout and a lost opportunity for dry eye relief. The Physician's evaluation of Restasis satisfaction in second trial (PERSIST) of topical cyclosporine ophthalmic emulsion 0.05% for dry eye study⁴⁸ was a multicenter, retrospective chart review that evaluated clinical outcomes of 35 patients with dry eye who received a second trial of CsA following a prior treatment failure, defined as discontinuation of CsA within 12 weeks. The second trial success was linked to patient education directly by physician provided in 97.1% of cases, and simultaneous topical corticosteroids in 28.6% of cases. Upon study conclusion, physicians reported on a questionnaire that 80% of patients benefited from a second trial of

Clinical Ophthalmology 2017:11

Table I Key papers: CsA for symptomatic relief in chronic dry eye

| / h | | | | | |
|---------------------------|---|----------|-------------------------|----------------------|---|
| Study | Title | No. of | Population | Treatment | Conclusion |
| | | patients | | | |
| Stevenson | Efficacy and safety of cyclosporine A ophthalmic emulsion | 162 | Moderate-to-severe | CsA 0.05%, 0.1%, | CsA 0.05% produced the most consistent |
| et al ²⁷ | in the treatment of moderate-to-severe dry eye disease: | | dry eye with or without | 0.2%, and 0.4%; | improvement in patient symptoms, but only the 0.2% |
| | a dose-ranging, randomized trial. The Cyclosporin A Phase II Study Group | | Sjögren's syndrome | versus vehicle | group had sustained significant improvement in OSDI |
| | | , | - | \(\delta \) | - 'c' |
| Sall et al" | I wo multicenter, randomized studies of the efficacy and | 1/9 | Moderate-to-severe | CsA 0.05% and 0.1%, | CsA 0.05% produced the most reliable subjective |
| | safety of cyclosporine ophthalmic emulsion in moderate | | dry eye with or without | versus venicle | improvement and required less rescue A l |
| | to severe dry eye disease. CsA Phase III Study Group | | Sjögren's syndrome | | |
| Stonecipher | The impact of topical cyclosporine A emulsion 0.05% on | 5,884 | Dry eye diagnosed by | CsA 0.05% | CsA significantly improved symptom severity by 30% |
| et al ⁴⁵ | the outcomes of patients with keratoconjunctivitis sicca | | the healthcare provider | | and activity impairment by 31% to 36%; 41% had |
| | | | | | rapid symptomatic relief response (within 1–3 weeks) |
| Wilson and | Long-term resolution of chronic dry eye symptoms and | 80 | Chronic dry eye | CsA 0.05% | Eight patients studied were cured for at least I year |
| Perry ⁴⁷ | signs after topical cyclosporine treatment | | | | after CsA; CsA may prevent progression of dry eye |
| | | | | | in some patients |
| Perry et al ²⁹ | Evaluation of topical cyclosporine for the treatment of | 158 | Mild, moderate, and | CsA 0.05% | Greatest improvement in OSDI occurred in mild |
| | dry eye disease | | severe dry eye | | disease |
| Rao ³⁷ | Topical cyclosporine 0.05% for the prevention of dry eye | 28 | Moderate-to-severe | CsA 0.05% versus | OSDI scores were significantly better in the |
| | disease progression | | dry eye | vehicle | CsA group |
| Chen et al ³⁴ | A comparison of cyclosporine 0.05% ophthalmic emulsion | 233 | Moderate-to-severe | CsA 0.05% versus | CsA superiority in symptomatic relief |
| | versus vehicle in Chinese patients with moderate to | | dry eye | vehicle | |
| | severe dry eye disease: an eight-week, multicenter, | | | | |
| | randomized, double-blind, parallel-group trial | | | | |
| Rao ⁴⁶ | Reversibility of dry eye deceleration after topical | 48 | Moderate-to-severe | CsA 0.05% versus AT | Earlier CsA treatment resulted in significantly better |
| | cyclosporine 0.05% withdrawal | | dry eye | | OSDI; maintenance CsA therapy reduced disease progression |
| Prabhasawat | A randomized double-masked study of 0.05% | 70 | Meibomian gland | CsA 0.05% versus AT | No difference in OSDI |
| et al ⁵⁰ | cyclosporine ophthalmic emulsion in the treatment of | | dysfunction dry eye | | |
| | meibomian gland dysfunction | | | | |
| Mah et al⁴8 | PERSIST: Physician's evaluation of Restasis $^{\scriptscriptstyle 	ext{@}}$ satisfaction in | 35 | Chronic dry eye | CsA 0.05% | The second trial success was linked to patient |
| | second trial of topical cyclosporine ophthalmic emulsion | | | | education directly by the physician (97.1%), and |
| | 0.05% for dry eye: a retrospective review | | | | simultaneous topical corticosteroid use (28.6%) |
| Sheppard | Effect of loteprednol etabonate 0.5% on initiation of dry | 112 | Mild-to-moderate | Loteprednol | Loteprednol etabonate group showed superior |
| et al ⁴⁹ | eye treatment with topical cyclosporine 0.05% | | dry eye | etabonate 0.5% or AT | results in OSDI improvement |
| | | | | 2000 | |

Abbreviations: AT, artificial tears; CsA, cyclosporine; OSDI, ocular surface disease index.

CsA, measured as improved patient symptoms in 57.1% and a reduction in corneal staining in 22.9%. The role of topical corticosteroids and CsA was studied by Sheppard et al⁴⁹ in a multicenter, randomized, double-masked parallel group, prospective study to compare pretreatment with loteprednol etabonate 0.5% (LE) to that with artificial tears prior to chronic CsA treatment. Hundred and twelve patients with mild-to-moderate DED were included. The LE group showed superior results in OSDI improvement. They found that LE induction improved patient tolerance and acceptance of chronic CsA therapy by addressing the CsA side effect of instillation burning.

CsA in the Asian population

Dry eye is particularly prevalent in the Asian population. Byun et al³⁹ performed a large prospective, multicenter, open-label surveillance study of 362 Korean patients with moderate-to-severe DED. After 3 months of CsA treatment, symptom scores and use of supplemental artificial tears were significantly reduced. Another smaller Korean study found less definitive results. Park et al⁴² compared CsA 0.05% emulsion to sodium hyaluronate 0.1%, 0.15%, or 0.3% in the treatment of 176 patients with non-SS, SS, and evaporative dry eye. They found that CsA had better, but not statistically significant, OSDI scores compared with the sodium hyaluronate groups. Similarly, outside of the Korean population, Prabhasawat et al⁵⁰ randomized 70 Thai patients in a prospective, double-masked, parallel-group clinical trial comparing CsA 0.05% bid with preservative-free artificial tears in patients with meibomian gland dysfunction for 3 months and found no difference between the two groups with respect to OSDI. In a small study of 38 female SS patients, Hyon et al⁵¹ demonstrated that CsA 0.05%, prednisolone acetate 1%, and autologous serum, in combination or when sequentially used, improved subjective symptoms, visual acuity, and fluorescein staining, but had a limited effect on Schirmer scores, tear breakup time, and rose bengal staining. Another example of mismatched subjective and objective findings occurred in a Turkish study⁵² that included 40 patients with SS and non-SS dry eye. They found OSDI score improvement in all the severities of dry eye, but only the mild-to-moderate stages had significant response in Schirmer score and tear breakup time.

CsA dosages

The majority of studies look at CsA emulsion, but Baiza-Duran et al⁵³ compared two different concentrations (0.05% and 0.1%) of CsA in aqueous solution to vehicle in moderate-to-severe DED. This was a multicenter, randomized, double-masked, vehicle-controlled trial of 183 Mexican patients which found CsA 0.1% in aqueous solution to be superior to the other treatment arms in the reduction of certain symptoms such as ocular dryness, red eye, photophobia, and ocular fatigue, but not for tearing or foreign body sensation. Again, discordantly, there was no difference in tear breakup time, corneal staining, and Schirmer scores when comparing CsA and vehicle.⁵⁴

Some patients who fail traditional on-label CsA 0.05% bid therapy may require more frequent application. Dastjerdi et al⁵⁵ performed a retrospective review of 22 patients with severe DED who after an inadequate response to a 4-month course of traditional bid CsA dosing were treated with off-label use of 3-4 times a day CsA. They found a significant response to this higher frequency dosing, as measured by symptom improvement and corneal staining. Although Phase II and III clinical trials established safety of higher concentrations of CsA, they did not demonstrate a definite dose-response relationship; therefore, the FDA approval was set at bid dosing. This study reported safety and an enhanced response in 68.2% of the 22 patients with ocular GVHD or SS. The group hypothesized that bid dosing was inadequate for this subset of severe DED patients because of: an incomplete drug effect due to insufficient dosing; endstage lacrimal gland disease or conjunctival scarring; and other individual risk factors and immunologic mechanisms not yet known.

Alternatively, some patients may tolerate a reduction of CsA 0.05% to qd frequency after 12 months of bid frequency. Su et al⁵⁶ randomized 50 patients to qd dosing and 50 patients to bid dosing; all of the patients had previously completed 12 months at bid dosing. At 6 months, there was no significant difference between the groups with regard to any outcome measure other than OSDI: surprisingly, the oncedaily group had superior results compared to the bid group.

CsA versus punctal occlusion versus combination

Roberts et al⁴⁰ randomized 30 patients into one of three treatment groups for 6 months: CsA 0.05% bid; lower lid punctal plugs; or plugs and CsA combination. They found that all the three methods treated dry eye, but the combination group produced the greatest reduction (5.5–6 to 1.6 times/d) in patient self-medication with artificial tears, a surrogate marker for relief from dryness and burning. Furthermore, continued reduction in lubricant use was noted during the course of the study in the combination group. This diminished

Clinical Ophthalmology 2017:11 submit your manuscript | www.dovepress.com

need for supplemental artificial tears was also observed by Sall et al.³² In contrast, Perry et al²⁹ did not find any difference in CsA response with punctal plugs.

CsA versus topical vitamin A

Kim et al³⁶ studied 150 Korean patients with DED and divided them into groups receiving CsA 0.05% emulsion bid, retinyl palmitate 0.05% qid, or only preservative-free artificial tears qid. They found that both CsA and vitamin A led to significant improvement in blurred vision when compared to the lubricant only group.

CsA in other autoimmune diseases, such as thryoid eye disease- and Stevens-Johnson syndrome-related chronic DED

There have been small studies on CsA in dry eye related to other systemic diseases. A Thai study⁵⁷ examined 17 patients with Stevens–Johnson syndrome for 6 months and found significant alleviation of symptoms including foreign body sensation, photophobia, eye pain, and dry sensation, which were graded by the visual analog scale. A Turkish study of 12 consecutive thyroid orbitopathy patients with dry eye showed a significant improvement in OSDI with 2 months of twice-daily CsA treatment.⁵⁸

CsA in trachomatous DED

Guzey et al³⁸ examined 64 patients with severe trachomatous DED and divided them into CsA 0.05% emulsion bid with nonpreserved artificial tears 5 times daily, and vehicle emulsion bid with nonpreserved artificial tears 5 times daily. They found that the CsA group had significantly improved OSDI.

CsA in hematopoietic stem cell transplantation patients

There are a handful of studies on the use of CsA in hematopoietic stem cell transplantation (HSCT) patients. The earliest study of CsA's potential benefit in DED related to GVHD was reported by Kiang et al⁵⁹ in 1998. They reviewed five case reports where topical CsA 1% was applied from twice daily to 8 times a day with varying results. This was followed by Lelli et al⁶⁰ who performed a retrospective chart review of 16 patients treated with CsA (varying concentrations/dosing) to validate safety. Rao and Rao⁶¹ undertook a 3-month prospective study of 8 patients who failed lubricant therapy, and found that CsA 0.05% bid for 3 months significantly improved Schirmer scores and tear breakup time. Subjective complaints of tearing, burning, and blurring were

also alleviated, but did not achieve statistical significance. Wang et al⁶² examined 13 patients with chronic GVHD in a prospective comparative study with seven patients in the CsA 0.05% emulsion qid group and six patients as a control group. There was a significant improvement in the visual analog scale symptom scores in the CsA group.

A further rationale for early treatment with CsA was supported by Malta et al.⁶³ They studied 105 patients in a retrospective, comparative, interventional case series where 81 patients received CsA 1 month prior to bone marrow transplant (BMT) and 24 patients did not receive CsA until at least 6 months after BMT (control group). Dry eye symptoms by OSDI were significantly more severe in the control group, even at the 2-year follow-up. They concluded that pre-BMT treatment with CsA reduces the lacrimal gland inflammation responsible for DED in BMT patients.

Boynton et al⁶⁴ studied 75 patients prior to initiation of HSCT, comparing CsA 0.05% bid to LE 0.5% bid in the prevention and progression of GVHD-related DED. This was a single-center, randomized, control group, prospective study with unclear masking that concluded LE to be safe and as effective as CsA in the prophylaxis and treatment of GVHD-related DED. Both groups of patients started either LE or CsA 1 month prior to HSCT. Unfortunately, despite the prophylactic therapy, the majority of patients in each group progressed to dry eye, and a minority in each group experienced improved OSDI and corneal staining by 12 months post-HSCT.

CsA in children with radiationassociated DED

Hoehn et al⁶⁵ did a retrospective review of 11 children who had radiation-associated DED that failed conventional therapy for at least 6 months. Symptomatic relief from CsA 0.05% emulsion bid was measured by intensity of photophobia; only three of the 11 children had any improvement. Although this study was limited, CsA 0.05% had little benefit on photophobia and corneal staining.

CsA in CL wearers

CL intolerance is often caused by dry eye, as lens wear can be associated with tear film alterations. ⁶⁶ Despite reducing wear time, using rewetting drops and lubricants, and managing meibomian gland dysfunction, some patients are unable to tolerate the lenses and discontinue their use. Hom⁶⁷ randomized 17 CL wearers in an investigator-masked study comparing twice-daily CsA 0.05% to rewetting drops for 5 weeks in patients with self-reported CL-related dryness. In the CsA group,

he noted a statistically significant increase in CL wear time and reduced amount of supplemental rewetting drops; OSDI scores improved, but there was not a statistically significant difference between the two groups. Conversely, Willen et al⁶⁸ studied 44 patients in a double-masked, randomized trial and found no significant difference between CsA and rewetting drop groups in objective or symptomatic parameters. The inability to show statistically significant improvement with CsA may be linked to the small sample sizes of these CL studies. Another consideration is that CsA may not be as effective in this subset of patients with dry eye.

CsA in laser vision correction patients

Laser in situ keratomileusis (LASIK) is associated with DED by exacerbating the symptoms in preexisting dry eye or by triggering it in previously asymptomatic patients. It is one of the most common complaints leading to patient dissatisfaction with the procedure.⁶⁹ Severing corneal nerves in the creation of the LASIK flap interrupts the neural feedback loop to the lacrimal glands with resultant diminished tear production.⁷⁰ In a small study of 21 patients with dry eye, Salib et al⁷¹ found that those patients treated with twice-daily CsA 0.05% 1 month prior to surgery had superior refractive predictability at 3 and 6 months after LASIK compared to those treated with artificial tears. However, OSDI scores were not statistically different between the groups. In a consecutive case series, Torricelli et al⁷² showed success with CsA 0.05% bid treating postoperative dry eye following LASIK and photorefractive keratectomy. After 12 months or more of CsA treatment, 5/82 (6.1%) eyes continued to have signs or symptoms of dry eye. Lee et al73 performed a retrospective, nonrandomized comparative analysis in 40 patients who underwent myopic laser epithelial keratomileusis and were treated with CsA 0.05% plus the conventional regimen of moxifloxacin 0.5%, fluorometholone 0.1%, and preservativefree sodium hyaluronate 0.1% qid; and 20 patients who received the conventional regimen alone. The CsA group had less ocular discomfort by 4 weeks postoperatively, especially those with preoperative dry eye. Likewise, Ursea et al⁷⁴ found faster recovery of vision in patients treated with CsA 0.05%, even at 1 week postoperatively. Conversely, in a study of 124 refractive patients, Hessert et al⁷⁵ found no difference in symptoms or visual recovery with or without CsA 0.05% bid for 3 months postoperatively.

Conclusion

The evidence for objective parameter response to CsA 0.05% bid in DED is strong, but subjective results are less

consistent. The lack of correlation in signs and symptoms is an inherent challenge of the disease.⁷ In addition, there is now a growing body of literature studying neuropathic ocular pain, an entity where mismatch in signs and symptoms is common, overlap with DED exists, and response to topical therapy is less effective.¹⁰ The current practice of symptom evaluation, slit-lamp examination, and metrics such as osmolarity may fail to delineate neuropathic ocular pain.⁷⁶

Subjective questionnaires that document symptoms and their impact on vision-related functioning, such as the OSDI, are widely used, but have limitations. Consisting of 12 questions, the OSDI evaluates for ocular symptoms of DED, limitations while performing vision-related activities, and environmental triggers. The answers are divided into five categories, ranging from "none of the time" to "all of the time." The scores range from 0 to 100 points, with 100 being the most severe DED.⁴⁴ Although the OSDI has been shown to be a valid and reliable tool in DED evaluation, the limited number of symptoms, activities, and triggers queried is a weakness. For instance, "sensitive to light," "gritty," "painful or sore," and "blurred or poor vision" are the only symptoms evaluated. Furthermore, the OSDI provides one total score to classify disease severity, but individual questions actually address different root causes of DED. For example, the questions regarding pain triggered by wind and light may indicate a neuropathic etiology of DED. 10,77

Future directions for investigation may include different formulations such as the 0.1% CsA cationic emulsion⁷⁸ Ikervis® (Santen SAS, Evry, France), recently launched in Europe, which is once-daily dosing; and novel delivery systems of CsA such as episcleral implant,79 ophthalmic insert,30 CL,80 and punctal plug.81 With the recent addition of lifitegrast, new studies will likely compare it with CsA and possibly evaluate it as a combination therapy. Further studies on potential synergism between LE and CsA may be warranted.⁴⁹ Developing newer questionnaires and enhancing analysis of current questionnaires may more accurately reflect the heterogeneity of DED, and improve our knowledge of this complex disorder. Despite the variability and limitations of study design and study end points, in general, CsA ophthalmic emulsion 0.05% bid appears to have utility in symptomatic relief in people with chronic dry eye.

Disclosure

FSM received grants from Allergan and is a consultant and a member of the speakers bureau of Allergan and Shire. The authors report no other conflicts of interest in this work. Rhee and Mah Dovepress

References

- Pflugfelder SC, Corrales RM, de Paiva CS. T helper cytokines in dry eye disease. Exp Eye Res. 2013;117:118–125.
- Restasis* (cyclosporine ophthalmic emulsion) [prescribing information].
 Irvine, CA: Allergan, Inc; 2014.
- Pflugfelder SC, Stern M, Zhang S, Shojaei A. LFA-1/ICAM-1 interaction as a therapeutic target in dry eye disease. *J Ocul Pharmacol Ther*. 2017;33(1):5–12.
- O'Brien, Collum LM. Dry eye: diagnosis and current treatment strategies. Curr Allergy Asthma Rep. 2004;4(4):314–319.
- Bron AJ, Tomlinson A, Foulks GN, et al. Rethinking dry eye disease: a perspective on clinical implications. *Ocul Surf.* 2014;12(2 Suppl): S1–S31.
- Sullivan BD, Crews LA, Messmer EM, et al. Correlations between commonly used objective signs and symptoms for the diagnosis of dry eye disease: clinical implications. *Acta Ophthalmol*. 2014;92(2): 161–166.
- Nichols KK, Nichols JJ, Mitchell GL. The lack of association between signs and symptoms in patients with dry eye disease. *Cornea*. 2004;23(8):762–770.
- 8. Vehof J, Smitt-Kamminga NS, Nibourg SA, Hammond CJ. Predictors of discordance between symptoms and signs in dry eye disease. *Ophthalmology*. 2017;124:280–286.
- 9. Rosenthal P, Baran I, Jacobs DS. Corneal pain without stain: is it real? Ocul Surf. 2009;7:28–40.
- Galor A, Levitt RC, Felix ER, Martin ER, Sarantopoulos CD. Neuropathic ocular pain: an important yet underevaluated feature of dry eye. Eye. 2015;29(3):301–312.
- [No authors listed]. The definition and classification of dry eye disease: report of the Definition and Classification Subcommittee of the International Dry Eye WorkShop (2007). *Ocul Surf.* 2007;5(2): 75–92
- Schein OD, Munoz B, Tielsch JM, Bandeen-Roche K, West S. Prevalence of dry eye among the elderly. *Am J Ophthalmol*. 1997;124(6): 723–728.
- Moss SE, Klein R, Klein BE. Prevalence of and risk factors for dry eye syndrome. Arch Ophthalmol. 2000;118(9):1264–1268.
- Schiffman RM, Walt JG, Jacobsen G, Doyle JJ, Lebovics G, Sumner W. Utility assessment among patients with dry eye disease. *Ophthalmology*. 2003;110(7):1412–1419.
- Uchino M, Schaumberg DA. Dry eye disease: impact on quality of life and vision. Curr Ophthalmol Rep. 2013;1(2):51–57.
- Dogru M, Tsubota K. Pharmacotherapy of dry eye. Expert Opin Pharmacother. 2011;12(3):325–334.
- Ridder WH 3rd, Tomlinson A, Huang JF, Li J. Impaired visual performance in patients with dry eye. *Ocul Surf.* 2011;9(1):42–55.
- Mertzanis P, Abetz L, Rajagopalan K, et al. The relative burden of dry eye in patients' lives: comparisons to a US normative sample. *Invest Ophthalmol Vis Sci.* 2005;46:46–50.
- Miljanovic B, Dana R, Sullivan DA, Schaumberg DA. Impact of dry eye syndrome on vision-related quality of life. *Am J Ophthalmol*. 2007; 143(3):409–415.
- Harmon D, Murphy R. Dry eye market set for robust growth. MarketScope. 2003:8:7–8.
- Galor A, Zheng DD, Arheart KL, et al. Dry eye medication use and expenditures: data from the medical expenditure panel survey 2001 to 2006. Cornea. 2012;31(12):1403–1407.
- Lam BL, Zheng DD, Davila EP, et al. Trends in glaucoma medication expenditure: medical expenditure panel survey 2001–2006. Arch Ophthalmol. 2011;129(10):1345–1350.
- Martin BI, Turner JA, Mirza SK, Lee MJ, Comstock BA, Deyo RA. Trends in health care expenditures, utilization, and health status among US adults with spine problems, 1997–2006. Spine. 2009;34(19): 2077–2084.
- Foulks GN, Forstot SL, Donshik PC, et al. Clinical guidelines for management of dry eye associated with Sjögren disease. *Ocul Surf.* 2015; 13(2):118–132.

 Shin MS, Kim JI, Lee MS, et al. Acupuncture for treating dry eye: randomized placebo-controlled trial. *Acta Ophthalmol*. 2010;88(8): e328–e333.

- McGhee CN, Dean S, Danesh-Meyer H. Locally administered ocular corticosteroids: benefits and risks. *Drug Saf.* 2002;25(1):33–55.
- Stevenson D, Tauber J, Reis BL. Efficacy and safety of cyclosporin A ophthalmic emulsion in the treatment of moderate-to-severe dry eye disease: a dose-ranging, randomized trial. The Cyclosporin A Phase 2 Study Group. *Ophthalmology*. 2000;107(5):967–974.
- Barber LD, Pflugfelder SC, Tauber J, Foulks GN. Phase III safety evaluation of cyclosporine 0.1% ophthalmic emulsion administered twice daily to dry eye disease patients for up to 3 years. *Ophthalmology*. 2005; 112(10):1790–1794.
- Perry HD, Solomon R, Donnenfeld ED, et al. Evaluation of topical cyclosporine for the treatment of dry eye disease. *Arch Ophthalmol*. 2008;126(8):1046–1050.
- Al-Saedi ZH, Alzhrani RM, Boddu SH. Formulation and in vitro evaluation of cyclosporine-A inserts prepared using hydroxypropyl methylcellulose for treating dry eye disease. *J Ocul Pharmacol Ther*. 2016;32(7):451–462.
- Stevenson W, Chauhan SK, Dana R. Dry eye disease: an immune-mediated ocular surface disorder. Arch Ophthalmol. 2012;130(1):90–100.
- Sall K, Stevenson OD, Mundorf TK, Reis BL. Two multicenter, randomized studies of the efficacy and safety of cyclosporine ophthalmic emulsion in moderate to severe dry eye disease. CsA Phase 3 Study Group. *Ophthalmology*. 2000;107(4):631–639.
- Kunert KS, Tisdale AS, Gipson IK. Goblet cell numbers and epithelial proliferation in the conjunctiva of patients with dry eye syndrome treated with cyclosporine. *Arch Ophthalmol*. 2002;120(3):330–337.
- 34. Chen M, Gong L, Sun X, et al. A comparison of cyclosporine 0.05% ophthalmic emulsion versus vehicle in Chinese patients with moderate to severe dry eye disease: an eight-week, multicenter, randomized, double-blind, parallel-group trial. *J Ocul Pharmacol Ther*. 2010;26(4): 361–366.
- Demiryay E, Yaylali V, Cetin EN, Yildirim C. Effects of topical cyclosporine a plus artificial tears versus artificial tears treatment on conjunctival goblet cell density in dysfunctional tear syndrome. *Eye Contact Lens*. 2011;37(5):312–315.
- Kim EC, Choi JS, Joo CK. A comparison of vitamin a and cyclosporine a 0.05% eye drops for treatment of dry eye syndrome. Am J Ophthalmol. 2009;147(2):206–213.
- 37. Rao SN. Topical cyclosporine 0.05% for the prevention of dry eye disease progression. *J Ocul Pharmacol Ther*. 2010;26(2):157–164.
- Guzey M, Karaman SK, Satici A, Ozardali I, Sezer S, Bozkurt O. Efficacy of topical cyclosporine a in the treatment of severe trachomatous dry eye. Clin Exp Ophthalmol. 2009;37(6):541–549.
- Byun YS, Rho CR, Cho K, Choi JA, Na KS, Joo CK. Cyclosporine 0.05% ophthalmic emulsion for dry eye in Korea: a prospective, multicenter, open-label, surveillance study. *Korean J Ophthalmol*. 2011; 25(6):369–374.
- Roberts CW, Carniglia PE, Brazzo BG. Comparison of topical cyclosporine, punctal occlusion, and a combination for the treatment of dry eye. *Cornea*. 2007;26(7):805–809.
- Hardten DR, Brown MJ, Pham-Vang S. Evaluation of an isotonic tear in combination with topical cyclosporine for the treatment of ocular surface disease. Curr Med Res Opin. 2007;23(9):2083–2091.
- Park Y, Song JS, Choi CY, Yoon KC, Lee HK, Kim HS. A randomized multicenter study comparing 0.1%, 0.15%, 0.3% sodium hyaluronate with 0.05% cyclosporine in the treatment of dry eye. *J Ocul Pharmacol Ther*. 2017;33(2):66–72.
- 43. Stonecipher KG, Torkildsen GL, Ousler GW 3rd, et al. The IMPACT study: a prospective evaluation of the effects of cyclosporine ophthalmic emulsion 0.05% on ocular surface staining and visual performance in patients with dry eye. Clin Ophthalmol. 2016;10:887–895.
- Schiffman RM, Christianson MD, Jacobsen G, Hirsch JD, Reis BL. Reliability and validity of the ocular surface disease index. *Arch Ophthalmol*. 2000;118(5):615–621.

- Stonecipher K, Perry HD, Gross RH, Kerney DL. The impact of topical cyclosporine a emulsion 0.05% on the outcomes of patients with keratoconjunctivitis sicca. *Curr Med Res Opin*. 2005;21(7):1057–1063.
- 46. Rao SN. Reversibility of dry eye deceleration after topical cyclosporine 0.05% withdrawal. *J Ocul Pharmacol Ther*. 2011;27(6):603–609.
- Wilson SE, Perry HD. Long-term resolution of chronic dry eye symptoms and signs after topical cyclosporine treatment. *Ophthalmology*. 2007;114(1):76–79.
- 48. Mah F, Milner M, Yiu S, Donnenfeld E, Conway TM, Hollander DA. PERSIST: Physician's evaluation of Restasis® satisfaction in second trial of topical cyclosporine ophthalmic emulsion 0.05% for dry eye: a retrospective review. Clin Ophthalmol. 2012;6:1971–1976.
- Sheppard JD, Donnenfeld ED, Holland EJ, et al. Effect of loteprednol etabonate 0.5% on initiation of dry eye treatment with topical cyclosporine 0.05%. Eye Contact Lens. 2014;40(5):289–296.
- Prabhasawat P, Tesavibul N, Mahawong W. A randomized double-masked study of 0.05% cyclosporine ophthalmic emulsion in the treatment of meibomian gland dysfunction. *Cornea*. 2012;31(12):1386–1393.
- Hyon JY, Lee YJ, Yun PY. Management of ocular surface inflammation in Sjögren syndrome. *Cornea*. 2007;26(9 Suppl 1):S13–S15.
- 52. Yuksel B, Bozdag B, Acar M, Topaloglu E. Evaluation of the effect of topical cyclosporine a with impression cytology in dry eye patients. *Eur J Ophthalmol*. 2010;20(4):675–679.
- Baiza-Duran L, Medrano-Palafox J, Hernandez-Quintela E, Lozano-Alcazar J, Alaníz-de la O JF. A comparative clinical trial of the efficacy of two different aqueous solutions of cyclosporine for the treatment of moderate-to-severe dry eye syndrome. *Br J Ophthalmol*. 2010;94(10): 1312–1315.
- Ames P, Galor A. Cyclosporine ophthalmic emulsion for the treatment of dry eye: a review of the clinical evidence. *Clin Investig (Lond)*. 2015;5(3):267–285.
- Dastjerdi MH, Hamrah P, Dana R. High-frequency topical cyclosporine 0.05% in the treatment of severe dry eye refractory to twice-daily regimen. *Cornea*. 2009;28(10):1091–1096.
- Su MY, Perry HD, Barsam A, et al. The effect of decreasing the dosage of cyclosporine a 0.05% on dry eye disease after 1 year of twice-daily therapy. *Cornea*. 2011;30(10):1098–1104.
- Prabhasawat P, Tesavibul N, Karnchanachetanee C, Kasemson S. Efficacy of cyclosporine 0.05% eye drops in Stevens Johnson syndrome with chronic dry eye. *J Ocul Pharmacol Ther*. 2013;29(3): 372–377.
- Gurdal C, Genc I, Sarac O, et al. Topical cyclosporine in thyroid orbitopathy-related dry eye: clinical findings, conjunctival epithelial apoptosis, and MMP-9 expression. *Curr Eye Res.* 2010;35(9):771–777.
- Kiang E, Tesavibul N, Yee R, Kellaway J, Przepiorka D. The use of topical cyclosporine a in ocular graft-versus-host-disease. *Bone Marrow Transplant*. 1998;22(2):145–151.
- Lelli GJ Jr, Musch DC, Gupta A, Farjo QA, Nairus TM, Mian SI. Ophthalmic cyclosporine use in ocular GVHD. *Cornea*. 2006;25(6): 635–638.
- Rao SN, Rao RD. Efficacy of topical cyclosporine 0.05% in the treatment of dry eye associated with graft versus host disease. *Cornea*. 2006;25(6):674–678.
- Wang Y, Ogawa Y, Dogru M, et al. Ocular surface and tear functions after topical cyclosporine treatment in dry eye patients with chronic graftversus-host disease. *Bone Marrow Transplant*. 2008;41(3):293–302.
- Malta JB, Soong HK, Shtein RM, et al. Treatment of ocular graftversus-host disease with topical cyclosporine 0.05%. *Cornea*. 2010; 29(12):1392–1396.

- 64. Boynton GE, Raoof D, Niziol LM, Hussain M, Mian SI. Prospective randomized trial comparing efficacy of topical loteprednol etabonate 0.5% versus cyclosporine-A 0.05% for treatment of dry eye syndrome following hematopoietic stem cell transplantation. *Cornea*. 2015; 34(7):725–732.
- Hoehn ME, Kelly SR, Wilson MW, Walton RC. Cyclosporine 0.05% ophthalmic emulsion for the treatment of radiation-associated dry eye in children. *Pediatr Blood Cancer*. 2013;60(7):E35–E37.
- Foulks GN. What is dry eye and what does it mean to the contact lens wearer? Eye Contact Lens. 2003;29(1 Suppl):S96–S100.
- Hom MM. Use of cyclosporine 0.05% ophthalmic emulsion for contact lens-intolerant patients. Eye Contact Lens. 2006;32(2):109–111.
- Willen CM, McGwin G, Liu B, Owsley C, Rosenstiel C. Efficacy of cyclosporine 0.05% ophthalmic emulsion in contact lens wearers with dry eyes. *Eye Contact Lens*. 2008;34(1):43–45.
- Jabbur NS, Sakatani K, O'Brien TP. Survey of complications and recommendations for management in dissatisfied patients seeking a consultation after refractive surgery. *J Cataract Refract Surg.* 2004; 30(9):1867–1874.
- Donnenfeld ED, Solomon K, Perry HD, et al. The effect of hinge position on corneal sensation and dry eye after LASIK. *Ophthalmology*. 2003;110(5):1023–1029.
- Salib GM, McDonald MB, Smolek M. Safety and efficacy of cyclosporine 0.05% drops versus unpreserved artificial tears in dry-eye patients having laser in situ keratomileusis. *J Cataract Refract Surg*. 2006;32(5):772–778.
- Torricelli AA, Santhiago MR, Wilson SE. Topical cyclosporine a treatment in corneal refractive surgery and patients with dry eye. *J Refract Surg*. 2014;30(8):558–564.
- Lee HS, Jang JY, Lee SH, Im SK, Yoon KC. Clinical effectiveness of topical cyclosporine a 0.05% after laser epithelial keratomileusis. *Cornea*. 2013;32(7):e150–e155.
- Ursea R, Purcell TL, Tan BU, et al. The effect of cyclosporine A (Restasis) on recovery of visual acuity following LASIK. *J Refract Surg*, 2008;24:473–476.
- Hessert D, Tanzer D, Brunstetter T, Kaupp S, Murdoch D, Mirzaoff M. Topical cyclosporine a for postoperative photorefractive keratectomy and laser in situ keratomileusis. *J Cataract Refract Surg.* 2013;39(4): 539–547
- Goyal S, Hamrah P. Understanding neuropathic corneal pain-gaps and current therapeutic approaches. *Semin Ophthalmol*. 2016;31(1–2): 59–70.
- 77. Crane AM, Feuer W, Felix ER, et al. Evidence of central sensitisation in those with dry eye symptoms and neuropathic-like ocular pain complaints: incomplete response to topical anaesthesia and generalised heightened sensitivity to evoked pain. *Br J Ophthalmol*. Epub 2017 Jan 18.
- Leonardi A, Van Setten G, Amrane M, et al. Efficacy and safety of 0.1% cyclosporine a cationic emulsion in the treatment of severe dry eye disease: a multicenter randomized trial. *Eur J Ophthalmol*. 2016; 26(4):287–296.
- Kim H, Csaky KG, Gilger BC, et al. Preclinical evaluation of a novel episcleral cyclosporine implant for ocular graft-versus-host disease. *Invest Ophthalmol Vis Sci.* 2005;46(2):655–662.
- Kapoor Y, Chauhan A. Ophthalmic delivery of cyclosporine A from brij-97 microemulsion and surfactant-laden p-HEMA hydrogels. *Int J Pharm.* 2008;361(1–2):222–229.
- 81. Gupta C, Chauhan A. Ophthalmic delivery of cyclosporine a by punctal plugs. *J Control Release*. 2011;150(1):70–76.

Clinical Ophthalmology 2017:11

Rhee and Mah Dovepress

Clinical Ophthalmology

Publish your work in this journal

Clinical Ophthalmology is an international, peer-reviewed journal covering all subspecialties within ophthalmology. Key topics include: Optometry; Visual science; Pharmacology and drug therapy in eye diseases; Basic Sciences; Primary and Secondary eye care; Patient Safety and Quality of Care Improvements. This journal is indexed on

 $\textbf{Submit your manuscript here:} \ \texttt{http://www.dovepress.com/clinical-ophthalmology-journal}$

Dovepress

PubMed Central and CAS, and is the official journal of The Society of Clinical Ophthalmology (SCO). The manuscript management system is completely online and includes a very quick and fair peer-review system, which is all easy to use. Visit http://www.dovepress.com/testimonials.php to read real quotes from published authors.